Author's response to reviews

Title: A retrospective study of the incidence, clinical characteristics, phenotypic identification, and antimicrobial susceptibility of bacteremic isolates of Acinetobacter ursingii

Authors:

Chun-Hsiang Chiu (pipi10279@gmail.com)
Yi-Tzu Lee (s851009@yahoo.com.tw)
Yung-Chih Wang (wystwyst@gmail.com)
Ti Yin (eden@mail.ndmctsgh.edu.tw)
Shu-Chen Kuo (ludwigvantw@gmail.com)
Ya-Sung Yang (ysyoung4097@gmail.com)
Te-Li Chen (tecklayyy@gmail.com)
Jung-Chung Lin (jclin@ndmctsgh.edu.tw)
Fu-Der Wang (fdwang@vghtpe.gov.tw)
Chang-Phone Fung (cpfung@vghtpe.gov.tw)

Version: 3
Date: 22 September 2015

Author's response to reviews: see over
To the Editor,

We wish to re-submit the revised manuscript (A retrospective study of the incidence, clinical characteristics, identification, and antimicrobial susceptibility of bacteremic isolates of Acinetobacter ursingii). The invaluable comments and suggestions are highly appreciated. We have addressed the reviewers’ suggestions as shown below. The changes we made in the manuscript are highlighted. All the authors have read and agreed to the re-submitted version of the manuscript.

Response to Referee 1:

Minor essential revisions:

1. Line 149: the sentence structure is clumsy - two strains had smeared DNA and two additional strains had 93% similar patterns? This could use some clarification.
   
   # Response: The sentences were rephrased. (line 153-154)

2. Line 187, 189, 190: strains were resistant or had intermediate susceptibility (to correct the English grammar).

   # Response: Revised as suggested: resistant or had intermediate susceptibility (line 191,193,194)

Additional Editorial requests:

1) Please add all email addresses for all of the authors on the front title page.
   # Response: the requested information had been added.

2) Please change the title of the "Introduction" section to "Background"
   # Response: Revised as requests (line 63)

3) Please add a short "Conclusion" section at the end of your Discussion (use the final paragraph(s) if applicable)
   # Response: Revised as suggested (line 244-246)

4) Please remove the Figure titles from the images - Figures should just be the images themselves.
   # Response: Revised as suggested.

Best regards
Te-Li Chen, MD, PhD.
Institute of Clinical Medicine, School of Medicine, National Yang-Ming University, Taiwan
Email: tecklayy@gmail.com